Prehospital early warning scores for adults with suspected sepsis: the PHEWS observational cohort and decision-analytic modelling study

Steve Goodacre,^{1,2*} Laura Sutton,¹ Kate Ennis,¹ Ben Thomas,¹ Olivia Hawksworth,¹ Khurram Iftikhar,² Susan J Croft,² Gordon Fuller,² Simon Waterhouse,¹ Daniel Hind,¹ Matt Stevenson,¹ Mike J Bradburn,¹ Michael Smyth,³ Gavin D Perkins,³ Mark Millins,⁴ Andy Rosser,⁵ Jon Dickson⁶ and Matthew Wilson¹

¹School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK
²Emergency Department, Northern General Hospital, Sheffield, UK
³Warwick Clinical Trials Unit, University of Warwick, Coventry, UK
⁴Yorkshire Ambulance Service NHS Trust, Wakefield, UK
⁵West Midlands Ambulance Service University NHS Foundation Trust, Midlands, UK
⁶Academic Unit of Primary Medical Care, University of Sheffield, Sheffield, UK

*Corresponding author s.goodacre@sheffield.ac.uk

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/NDTY2403.

Primary conflicts of interest: Steve Goodacre reports chairing the NIHR Clinical Trials Unit Standing Advisory Committee and membership of the following groups and committees: HTA PCCPI Methods Group, HTA Prioritisation Committee A Methods Group, HTA Prioritisation Committee B Methods Group, HTA Post-Funding Committee, HTA Funding Committee Policy Group and HTA Commissioning Committee. Laura Sutton reports participation on a Data Monitoring and Ethics Committee for a NIHRfunded trial in an unrelated area of study. Susan J Croft reports receiving payments as an Emergency Medicine expert advisor for the Welsh Ombudsman and membership of the NICE sub-arachnoid haemorrhage committee as an Emergency Medicine advisor. Daniel Hind reports an NIHR HTA grant outside of the submitted work and membership of the HTA Fast Track Committee. Mike J Bradburn reports membership of the HTA Commissioning Committee. Michael Smyth reports NIHR grants outside of the submitted work. Gavin D Perkins reports grants from the NIHR and Laerdal Foundation outside of the submitted work, honorary membership of the Intensive Care Society and memberships of the following: NIHR HTA Clinical Effectiveness Board, NIHR Clinical Trials Advisory Committee, NIHR Academy, NIHR CTU Standing Advisor Committee, NIHR HTA Clinical Evaluation and Trials Committee and COVID-19 Reviewing Committee. He reports honoraria from Elsevier for editorial roles for Resuscitation and Resuscitation Plus journals. Matthew Wilson reports an NIHR Research for Patient Benefit grant outside of the submitted work, participation on a Data Monitoring Committee for an NIHR-funded trial in an unrelated area of study and the role of Grants Officer to the National Institute of Academic Anaesthesia.

Published March 2024 DOI: 10.3310/NDTY2403

Plain language summary

Prehospital early warning scores for adults with suspected sepsis: the PHEWS observational cohort and decision-analytic modelling study

Health Technology Assessment 2024; Vol. 28: No. 16 DOI: 10.3310/NDTY2403

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

Sepsis is a life-threatening condition in which an abnormal response to infection causes heart, lung or kidney failure. People with sepsis need urgent treatment. They need to be prioritised at the emergency department rather than waiting in the queue. Paramedics attempt to identify people with possible sepsis using an early warning score (based on simple measurements, such as blood pressure and heart rate) alongside their impression of the patient's diagnosis. They can then alert the hospital to assess the patient quickly. However, an inaccurate early warning score might miss cases of sepsis or unnecessarily prioritise people without sepsis. We aimed to measure how accurately early warning scores identified people with sepsis when used alongside paramedic diagnostic impression.

We collected data from 71,204 people that two ambulance services transported to four different hospitals in 2019. We recorded paramedic diagnostic impressions and calculated early warning scores for each patient. At one hospital, we linked ambulance records to hospital records and identified who had sepsis. We then calculated the accuracy of using the scores alongside diagnostic impression to diagnose sepsis. Finally, we used modelling to predict how many patients (with and without sepsis) paramedics would prioritise using different strategies based on early warning scores and diagnostic impression.

We found that none of the currently available early warning scores were ideal. When they were applied to all patients, they prioritised too many people. When they were only applied to patients whom the paramedics thought had infection, they missed many cases of sepsis. The NEWS2, score, which ambulance services already use, was as good as or better than all the other scores we studied. We found that using the NEWS2, score in people with a paramedic impression of infection could achieve a reasonable balance between prioritising too many patients and avoiding missing patients with sepsis.

Health Technology Assessment

ISSN 2046-4924 (Online)

Impact factor: 3.6

A list of Journals Library editors can be found on the NIHR Journals Library website

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2022 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), EMBASE (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as award number 17/136/10. The contractual start date was in July 2019. The draft report began editorial review in April 2023 and was accepted for publication in August 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2024 Goodacre *et al.* This work was produced by Goodacre *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).